Author's response to reviews

Title: A prospective, randomized, double-blind, and multicenter trial of prophylactic effects of ramosetron on postoperative nausea and vomiting (PONV) after craniotomy: comparison with ondansetron

Authors:

Jung-Hee Ryu (jinaryu@lycos.co.kr)
Ji-Eun Lee (lively12@gmail.com)
Young Jin Lim (limyjin@snu.ac.kr)
Hee-Pyoung Park (hppark@snu.ac.kr)
Deok Man Hong (mellyn7@gmail.com)
Hee Jung Baik (baikhj@ewha.ac.kr)
Kyeong Tae Min (ktmin501@yuhs.ac)
Hyunzu Kim (aneshyunzu@yuhs.ac)
Jong In Han (hanji@ewha.ac.kr)
Sang-Hwan Do (shdo@snu.ac.kr)

Version: 6
Date: 24 July 2014

Author's response to reviews:

The Editorial Office
BMC anesthesiology
July 24, 2014

A prospective, randomized, double-blind, and multicenter trial of prophylactic effects of ramosetron on postoperative nausea and vomiting (PONV) after craniotomy: comparison with ondansetron

Dear Editor

It has not been submitted for publication nor has it been published in whole or in part elsewhere. We attest to the fact that all authors listed in the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission to BMC anesthesiology. No person or persons other than the authors listed have contributed significantly to its preparation.

Most of all, we would like to thank you for the careful and thoughtful suggestions on our manuscript. We have revised the manuscript based on these suggestions. As a result, we believe that our manuscript is greatly improved.

Sincerely,

Jung-Hee Ryu, M.D. Ji-Eun Lee, M.D. Young Jin Lim, M.D. Hee-Pyoung Park, M.D. Deok Man Hong, M.D. Hee Jung Baik, M.D. Kyeong Tae Min, M.D. Hyunzu Kim, M.D. Jong In Han, M.D. Sang-Hwan Do, M.D.
* Corresponding author: Sang-Hwan Do
Department of Anesthesiology & Pain Medicine
Seoul National University, College of Medicine
Seoul National University Bundang Hospital
166 Gumi-ro, Bundang-gu, Seongnam-si
Gyeonggi-do 463-707, South Korea
Tel: 82-31-787-7501
Fax: 82-31-787-4063
E-mail: shdo@snu.ac.kr
Trial registration: CRiS, registration number KCT0000320
Category: Original Research

Editor's Comment:
We note that your tables have been uploaded as additional files but are not referred to in the text as such. As additional files they will not be displayed in the text. Please remove the tables as additional files and incorporate them into the manuscript, after the references, OR amend the text (e.g. page 1, last paragraph) to read additional file 1 instead of table 1. Please see the following link for details on formatting these files:
http://www.biomedcentral.com/bmcanesthesiol/authors/instructions/researcharticle
: The authors appreciate for your advice. Tables are incorporated to the main text after references (Page 25-29).